Author's response to reviews

Title: Cyclical changes in seroprevalence of leptospirosis in California sea lions: endemic and epidemic disease in one host species?

Authors:

James O Lloyd-Smith (jlloydsmith@psu.edu)
Denise J Greig (greigd@tmmc.org)
Sharon Hietala (skhietala@ucdavis.edu)
George S Ghneim (George.Ghneim@ncmail.net)
Lauren Palmer (lpalmer333@aol.com)
Judy St. Leger (Judy.St.Leger@seaworld.com)
Bryan T Grenfell (grenfell@psu.edu)
Frances M D Gulland (gullandf@tmmc.org)

Version: 6 Date: 2 November 2007

Author's response to reviews:

This is a final version of the manuscript, with formatting changes as requested in your email of Nov 2, 2007.